Author's response to reviews

Title: Higher incidence of perineal community acquired MRSA infections among toddlers

Authors:

Alexis C Mccullough (Alexis.Mccullough@utoledo.edu)
Melissa Seifried (Melissa.Seifried@gmail.com)
Xiaochen Zhao (xzhao24@buffalo.edu)
Jeffrey Haase (jeffrey_haase@mhsnr.org)
William J Kabat (bkabat@childrensmemorial.org)
Ram Yogev (RYogev@childrensmemorial.org)
Robert M Blumenthal (Robert.Blumenthal@utoledo.edu)
Deepa Mukundan (Deepa.Mukundan@utoledo.edu)

Version: 8 Date: 25 October 2011

Author's response to reviews:

Page 5 under the heading "Discussion"

paragraph 3 the references [25-27] is changed to [25-26,28] and in the last paragraph the (Table 4 and Figure 2) is be changed to (Table 3 and Figure 2).

Page 7 under the heading "Abbreviations" the paragraph should end in a full stop instead of a comma.

Page 11

Figure 2 should read Tables 2, 3 and 4 instead of Table 4.

Page 12

Last row with PFT subtype was deleted in the table. The same information is there is Table 3.